Author’s response to reviews

Title: Effectiveness of text messages for decreasing inactive behaviour in patients with knee osteoarthritis: a pilot randomised controlled study

Authors:
Cecilie Bartholdy (cecilie.roedgaard.bartholdy@regionh.dk)
Henning Bliddal (henning.bliddal@regionh.dk)
Marius Henriksen (marius.henriksen@regionh.dk)

Version: 2 Date: 26 Apr 2019

Author’s response to reviews:
We thank you for considering our paper for publication with minor revisions.

COMMENT, REVIEWER REPORTS

I would insist however, that you do not say that the results were not significant but rather to say that there were no meaningful differences between groups on activity measures. With an under powered study there could be differences that did not meet "significance", however, as you pointed out, there were no differences of any importance to detect.

AUTHOR RESPONSE

We agree and have change the wording of the results to say that there was no meaningful difference between the groups on activity measures.

AUTHOR ACTION

The word significant have been removed from the results and discussion.